Author's response to reviews

Title: Antenatal mindfulness intervention to reduce depression, anxiety and stress: a pilot randomised controlled trial of the MindBabyBody program in an Australian tertiary maternity hospital

Authors:

Hannah Woolhouse (hannah.woolhouse@mcri.edu.au)
Kristine Mercuri (Kristine.Mercuri@thewomens.org.au)
Fiona Judd (Fiona.Judd@thewomens.org.au)
Stephanie Brown (stephanie.brown@mcri.edu.au)

Version: 2  Date: 24 April 2014

Author's response to reviews:

1. Line numbering has been added to the manuscript as requested
2. Institutions and countries of all co-authors is included on the title page of the manuscript
3. Tables have now been included in the main manuscript following the references.